
Mauna Loa Solar Observatory Observer's Log

Sun Apr 1 17:22:12 GMT 2001

Year: 01 Doy: 091 Observer: koon

WEATHER COMMENT: Sun Apr 1 17:22:16 GMT 2001

Very thick cirrostratus overcast, probably no observing today.

GONG PROBLEM: Sun Apr 1 18:05:38 GMT 2001

Going down to the shelter to fix a tape drive problem.

GONG PROBLEM: Sun Apr 1 19:07:56 GMT 2001

OK, back from the shelter. We are awaiting 2 new tape drives for the

secondary tape bank. The 2 good drives are now in the primary tape bank. The instrument is back up.

WEATHER COMMENT: Sun Apr 1 20:24:42 GMT 2001

The cloud cover is getting thicker, no observing today.

Sun Apr 1 20:25:20 GMT 2001

MkIV